Review

Falls Incidence and Associated Risk Factors among People with Chronic Obstructive Pulmonary Disease (COPD)

Mehrdad Davoudi ^{1,2}, Seyyed Mohammadreza Shokouhyan ¹, Mohsen Abedi ³, Mehdi Rezaei ⁴, Mohamad Parnianpour ¹, Kinda Khalaf ^{5,*} and Masood Mazaheri ⁶

- ¹ Department of Mechanical Engineering, Sharif University of Technology, Tehran, Iran
- ² Clinic for Orthopaedics and Trauma Surgery, Heidelberg University Hospital, Heidelberg, Germany
- ³ Chronic Respiratory Diseases Research Center (CRDRC), School of Rehabilitation, Shahid Beheshti University of Medical Sciences, Tehran, Iran
- ⁴ Department of Physiotherapy, School of Rehabilitation, Shahid Beheshti University of Medical Sciences, Tehran, Iran
- ⁵ Healthcare Engineering Innovation Center, Department of Biomedical Engineering, Khalifa University of Science and Technology, Abu Dhabi, United Arab Emirates
- 6 Department of Plastic and Reconstructive Surgery, Erasmus MC Cancer Institute, University Medical Center Rotterdam, Rotterdam, The Netherlands

Abstract: Chronic obstructive pulmonary disease (COPD) is increasingly being recognized as a systemic disease rather than a mere disorder of the lungs. Central (respiratory) and peripheral (limb) muscle weakness are among the main pronounced systemic effects of COPD. While the disease primarily affects the lower limb muscles and contributes to gait impairment, COPD is also associated with an increasing risk of falls in patients (COPDp). Previous studies have reported higher rates of falls among COPDp (1.17 to 1.20 falls/person-year), amounting to four times higher than an agematched healthy group. Potential fall risk factors include muscle weakness, impaired daily activities, cognitive dysfunction, and gait and balance impairment. Although COPDp often manifest many of these risk factors, there remains a gap in literature regarding falls during walking in this population. This study aimed to 1. analyze the literature to identify the risk factors of falling in COPDp, and 2. investigate the underlying mechanisms by which these risk factors can lead to increased prevalence of falling. The results suggest that in addition to the known risk factors of falling, low back pain and mental fatigue should also be considered as relevant risk factors in the treatment process of these patients. Moreover, respiratory problems, which are common in this population, have demonstrated pronounced effects on energy expenditure, gait, and other types of activities of daily living (ADLs), leading to reduced intensity, disrupted coordination of the trunk-pelvic structure with the lower limbs during gait, and altered motor control performance due to activation of muscles in an inefficient synergic manner. These problems potentially lead to the increased vulnerability of these patients to external disturbances and higher incidence risk of falls and injuries. Cognitive problems, which are typically due to reduced oxygen received by the brain, as well as general inflammation caused by COPD, also play a significant role in gait disruption and balance. Future research is warranted to determine the prevalence of falls in COPDp by examining the response of these patients to Medio-Lateral (ML) and Anterior-Posterior (AP) disturbances during gait in association with traditional and recommended fall risk factors.

Keywords: chronic obstructive pulmonary disease; COPD; fall risk factor; gait; balance; cognition; daily activity; muscle dysfunction

1. Introduction

There is little debate that the Corona virus, with its unprecedented scale and severity, has changed human life and redirected human priorities for decades to come. The impact of COVID-19 has extended far beyond a health crisis with devastating

^{*}Corresponding author: kinda.khalaf@ku.ac.ae; Tel.: +971-(0)2-4018107

mortality/morbidity rates around the world to include unpredictable changes on the quality of social life and work environments, and unprecedented long-term global socioeconomic challenges. On the other hand, the pandemic has also created a test bed for the rapid emergence of disruptive innovations and technologies in many fields. Considering the potentially lethal effects of the virus on the lungs and the respiratory system, the field of rehabilitation has quickly evolved to define specific rehabilitation exercises tailored for Corona recovery (1), and to introduce novel physiological risk markers, such as the use of gait speed as a biomarker to assess the risk of contracting the virus (2). The impact of the Corona virus, as a severe lung disease, on the neuromuscular and cardiovascular systems among others, is an example of a cascade of systemic complications impacting several organ systems, suggesting the critical value of multifactorial comprehensive investigations.

COPD is a chronic lung disease, which also instigates systematic effects on other vital organs of the body, such as the neuromuscular system. Characterized by progressive and persistent expiratory air flow limitations (3), COPD results from long-term exposure to harmful gases and particles combined with individual factors, including events which influence childhood lung growth, such as severe asthma or bronchitis, and genetics (AAT deficiency). Unlike asthma, however, which is more common in children and adolescents, COPD is more common in the elderly (over 45 years) (4, 5). Prior to the recent pandemic, COPD was considered as the third cause of death in the world (6) with nearly 3.23 million deaths in 2019 (https://www.who.int/news-room/fact-sheets/detail/chronic-obstructive-pulmonary-disease-(copd)) and more than 300 million infections (50 million in Europe) (7-10). Environmental exposure to tobacco smoke, indoor air pollution and occupational dusts, fumes and chemicals are important risk factors for COPD, in addition to the increase in the phenomena of fine dusts and air pollution in big cities, as well as the aging of communities. Although COPD has traditionally been more prevalent among men (5), increasingly more women are being diagnosed with the disease in recent years, possibly due to the increasing number of women smokers around the world (11).

Moreover, while COPD is more typically associated with less developed countries, more than 16 million people (6.6% of adults) in the United States have been diagnosed with the disease, ensuing a huge financial burden of \$72 billion a year (12). In the EU, the financial cost is estimated at \$23.3 billion per year (10). For example, in Germany, the prevalence of COPD in people older than 18 is estimated at 5.8%, and increases to 18.1% for men older than 40 (13, 14). Therefore, although COPD has variable prevalence in different parts of the world, according to the geography, culture, economy, and environmental conditions, it is considered as a growing global health challenge, which is expected to continue to increase in prevalence and severity, along with global pandemics, the aging population and severe environmental changes associated with global warming (15).

The physical attributes of COPD include symptoms such as coughing, wheezing, and shortness of breath. COPD also leads to widespread inflammation throughout the body, which along with the challenges and frustration associated with chronic disease, could cause the reported cognitive problems presenting in these patients (16) (20). Moreover, COPDp often struggle with cardiovascular problems and other disabilities, as well as breathlessness or difficulty in breathing, which typically result in these patients leading a sedentary lifestyle (17) (18). This, in addition to sleeping difficulties (probably due to dyspnea), are among the main reasons for the increasing prevalence of depression and anxiety among COPDp and the increased intake of antidepressants or hypnotics (19). Constant health concerns, frustration/aggravation, lack of a definitive treatment and inability to perform many daily activities, all impact different aspects of the cognitive function of chronic patients. This is further exasperated in COPDp by the lack of proper oxygen delivery to the brain due to lung dysfunction and hence potential challenges at the brain level (21). This suggests that muscle problems in COPDp may not only be due to changes in muscle structure, (i.e. muscle atrophy), decreased oxygen uptake, as well as mass and CSA, but also due to an impaired neural drive (22), considering evidence that the problems of oxygen delivery to the prefrontal cortex reduce central motor drive (21). Moreover, Vitamin D deficiency, which has been frequently reported in COPD patients, also has been known to exacerbate musculoskeletal problems in these patients (23).

Although COPD is known as an internal lung-related disease with respiratory complications, as previously mentioned, non-respiratory manifestations of COPD, such as musculoskeletal problems and motor disorders significantly impact the lives of these patients, and may disrupt the treatment process and aggravate the disease (25, 26). Specifically, increased prevalence of accidental falls in COPDp is one of the systematic challenges associated with the disease which negatively impacts the quality of life and life expediency of COPDp (27). Potential fall risk factors include muscle weakness, impaired daily activities, cognitive dysfunction, and gait and balance impairment. Due to the relevance of the topic, numerous researchers investigated these factors although there remains a gap in literature regarding falls during walking in this population. Furthermore, while numerous studies reported motor-balance problems in COPD patients and their association with the increased risk of falling, the underlying mechanism leading to falls in COPDp remains elusive. The question of how much is inactivity, energy expenditure, gait difficulties, and cognitive problems responsible for COPD patients falling remains an open area of research. This scoping review aimed to 1. analyze the literature to identify the risk factors of falling in COPDp, and 2. investigate the underlying mechanisms by which these risk factors can lead to increased prevalence of falling. The remainder of this review is organized as follows: Section 2 describes the methods used for the review including the study approach, strategy, criteria, and data extraction. Section 3 describes falls in COPDp, while section 4 details the main intrinsic and extrinsic fall risk factors, including muscle dysfunction, impaired activities, balance and gait deficits, and cognitive functioning. Section 5 presents the limitations and section 7 highlights the conclusions and future directions.

2. Methods

2.1. Study Approach

A literature review was conducted to identify/summarize the fall risk factors in people with COPD towards elucidating possible mechanisms underlying the increased risk of falls in these individuals. Intrinsic risk factors such as muscle weakness, gait and balance deficits, impaired daily activity, and cognitive problems are included in this review, as well as low back pain and mental fatigue as other potential risk factors.

Considering the nature and intent of this study, we opted for a scoping rather than a systematic review. Scoping reviews typically allow for a broader overview of the area of focus and can tackle a broader research question. As we aimed to address both research and clinical communities, a scoping review was more suitable expanding the range of readers and allowing for a more extensive inclusion criterion (24).

2.2. Search strategy

A comprehensive literature search was conducted within Google Scholar, ScienceDirect, and PubMed databases using a combination of keywords from the following groups: (1) "COPD" or "Chronic Obstructive Pulmonary Disease" (2) "Fall" or "Fall Incidence" or "Fall Risk Factor" (3) "Muscle" or "Gait" or "Balance" or "Daily Activity" or "Cognition" or "Mental Fatigue" or "Low Back Pain". The search was limited to articles published in English before August 2022. The title and abstract of the search results were screened against the inclusion/exclusion criteria (next section). The full text of the relevant articles was reviewed and those meeting the study criteria were considered for further analysis. The references were checked to identify additional articles for possible inclusion in this study.

2.3. Study Selection Criteria

Articles were screened against the following inclusion/exclusion criteria. Articles for inclusion had to address (1) intrinsic and extrinsic fall risk factors in patients with COPD including muscle weakness, gait and balance deficit, impaired daily activity, and

cognitive problems (2) underlying mechanisms for increased the fall incidence in these patients. Precipitating fall risk factors such as exacerbations and dyspnea are outside the scope of this study and not included in the review. These factors are acute episodes related to COPD. Articles analyzing the effects of medication, visual deficit, use of an assistive device and nutritional depletion and malnutrition, were also excluded. The articles meeting the inclusion and exclusion criteria were screened for value provided in fall risk assessment in COPD patients.

2.4. Data Extraction (fill the initials here)

Following the screening of the full-text articles based on the criteria, a primary reviewer (MD) performed the data extraction. A part of the data extraction was also carried out by additional reviewers (MM, SMS). The results were verified by (MP, KK). The clinical application and relevance were discussed and confirmed by our clinical collaborators (MM, MA, MR). The data extraction focused on two broad areas: well-known fall risk factors related to the physical and psychological status of the COPD patient and the potential risk factors related to the possible causes of falls in these patients.

3. Falls in COPD patients

Although COPD is known as an internal lung-related disease with respiratory manifestations, such as shortness of breath, chest tightness, dry cough and phlegm, as previously mentioned, non-respiratory manifestations of COPD, such as musculoskeletal problems and motor disorders significantly impact the lives of these patients, and may disrupt the treatment process and aggravate the disease (25, 26). Increased prevalence of accidental falls in COPDp is one of the systematic challenges associated with the disease which negatively impacts the quality of life and life expediency in these people (27).

In a review article, Riog et al. (27) examined the risk factors associated with falls in COPDp and classified them into two risk groups: intrinsic risk factors, such as gait and balance deficits, impaired activity of daily living, depression and cognitive problems, and medication and precipitating (acute) risk factors including exacerbation and dyspnea. Two years later in another study (28) they surveyed a population of 101 patients over 6 months using various questionnaires, such as Activities Balance Confidence (ABC) Scale and Physical Activity Scale for the Elderly (PASE), as well as structured forms for recording demographic data, current medications, co-morbidities, oxygen use, acute exacerbations of COPD, falls during the preceding 6 months and use of an assistive device. The results indicated that fall occurrence was relatively independent of the stage of COPD (25). Thirty-two participants (31.7%) reported at least one fall in the past six months, which was more than the number of fallers reported for an age-matched elderly group (27%). In addition, the incidence of falls in the COPD population was 1.2 falls per person-year, which is approximately 4 times the number reported for the healthy elderly (0.24 falls per person-year). Taking more medications (8 drugs vs. 5 drugs), age (75.5 years vs. 72.1 years), gender (60% among women), coronary heart disease, and dyspnea were recorded as the most important causes of falls.

Beauchamp et al. conducted three separate studies on the relationship between falls and COPD (29-31). Out of the 39 subjects who participated in their study, 46% reported falls at least once a year (29), which was higher than the findings reported by Riog et al. (28). Although the fallers were slightly older (71 vs. 69 years), there was no significant difference between the gender of fallers and non-fallers. Moreover, no significant difference was seen between the data of FEV1 (forced expiratory volume in the first second) obtained from spirometry between the two groups, while the use of supplemental oxygen was an important predictor in diagnosing fallers (29). Examining the balance status of these patients, revealed that compared to age- and sex- matched healthy people in all aspects of balance measured by Berg Balance Scale (BBS), the COPDp had a lower score (29, 31). They also compared the balance status of the faller and non-faller groups of COPD individuals, which indicated that patients with a history of fall had more balance

problems (29). This study suggested that gait and balance impairments should be considered as risk factors for the fall incidence in these COPDp, and recommended that balance exercises should be prescribed during respiratory rehabilitation (31). This suggestion was the basis of a third study by the same group (30), in which they demonstrated that doing six weeks (three times a week) of balance exercises, along with breathing exercises, improved all the criteria of balance, lower body muscle strength and daily activity of patients as compared to the control group, who received only breathing exercises. This study confirmed that strengthening the lower body muscles and maintaining regular balance exercises are likely to reduce the risk of falling. This research group was the first to use a lean-and-release system to examine the differences between the response of COPDp and healthy individuals to unexpected perturbations (31). The importance of this work lies in the fact that falling in real life usually happens unexpectedly and is often triggered by external disturbances, such as slipping, stepping, or hitting an obstacle.

Simulating of such scenarios in a laboratory environment may help understand the governing mechanisms and underlying problems that lead to ineffective response leading to a fall. Beauchamp et al. (31) Reported on foot-contact times in response to delayed foot-off perturbations in COPD patients, indicating delayed response in unstable conditions, which is consistent with the longer anticipatory phase observed in these patients. In addition, although the onset of the lower body muscles of the COPD group was not significantly different from those of healthy individuals (except for the tibialis anterior muscle) in relation to the disturbance occurrence, the use of a dynamometer revealed that the torque produced by all lower torso muscles was significantly less than that in healthy individuals. This may explain not producing sufficient compensatory joint torque to restore balance in these patients, suggesting that the incorporation of balance exercises for patients with COPD should also include perturbation-based training (|28).

In a one-year study of 39 COPD patients in Australia, Oliveira et al. (32) reported a prevalence of 40% of fall incidence among these patients. They also identified the number of medications, comorbidities, and prior history of falls, as fall risk factors. Although smoking history (number of packs in a year) was an important risk factor among fallers, no correlation was found between lung function and fall incidence, which is also consistent with the findings of Riog et al. (28) regarding the independence of fall occurrence from the stage of COPD (obtained by spirometry). They cited the relationship between long-term smoking and balance performance among smokers to explain their work and suggested the relationship between balance problems and the occurrence of falls in COPD individuals as a probable mechanism. In their next study in 2017 (33), they reported 1.76 falls/person-year in COPD patients with acute exacerbation after hospital discharge, which was significantly higher as compared to the reported results for healthy elderly or COPD patients with stable conditions. This phenomenon was attributed to the deterioration of balance after discharge, accompanied by a decrease in quadriceps strength, increased dyspnea, as well as an increase in sway while standing. The Australian group's research has well demonstrated the importance of balance problems caused by COPD among people as a risk factor for falls. While the above studies showed the importance of falls as a common problem among COPD patients, most of these studies have included a limited number of patients (less than 100). Hakami et al. (34) used the UK primary care database to examine the incidence of falling among 44,400 COPD patients as compared with 175,545 healthy individuals. After adjusting for smoking status, age, gender and other confounding factors, patients with COPD were 47% more likely to have a record incidence of fall compared with non-COPD patients.

Interestingly, previous research has reported different results regarding the prevalence of falling in COPD individuals. A Swedish team in 2009 (35) investigated the fall occurrence in 80 COPD patients with an average age of 65, and reported 25 percent of the subjects had at least one fall per year. Another team from the University of KU Leuven in Belgium (36) studied a population of 44 people with COPD and reported a falling occurrence rate of 20%. They suggested that the occurrence of falling was seen only among very frail hospitalized patients, which is not comparable to other frail elderly patients. Indeed,

they considered falling in COPD as an age-related issue, and not a component of the systemic effects of the disease, which contrasts with more. Recent results published by Riog, Beauchamp, and Oliveir and their groups (27-33). In particular, a study by Hakami et al. (34), which examined a large COPDp of these patients, found that the risk of falling was significantly higher in these patients as compared with healthy elderly of the same age group. The contrasting results obtained by the Swedish and Belgian groups may reflect disparity in healthcare around the world in association with economic, cultural, and social aspects. Countries with quality, accessible and affordable healthcare systems provide people, especially the elderly, with effective preventive medicine and better control and treatment options, highly relevant for systemic disease. The effect of socioeconomic variations in chronic obstructive pulmonary disease prevalence, diagnosis, and treatment has been extensively investigated, with the results reflecting that 90% of all reported deaths caused by COPD occurs in low-income countries (37).

4. Main fall risk factors in COPD patients

The multiplicity of risk factors for the occurrence of falls in COPD patients include typical the fall risk factors seen in the elderly, such as aging, female gender, history of falling and increasing comorbidities, in addition to COPD-specific factors, such as balance and gait disorders, muscle weakness, depression and cognitive disorders, increase in the number of medications, use of supplemental oxygen, increased 'fear of falling' and heavy smoking history. This is confirmed by data revealing that the probability of falling in this population is approximately 50% higher than age- and gender- matched older people.

Falls are considered as one of the principal causes of injury and hospitalization among the elderly and people with COPD. Statistics indicate that COPDp are among the top reasons for hospital referrals (38). Hip fracture is one of the main reasons for the loss of independence in the elderly, reducing the quality of life, and also is associated with increased mortality, with mortality rates of 26% and 37% for men and women (39), respectively, after one year post hip fracture. In 90% of the time, these fractures occur due to falls and incur high economic costs for the individual and the health care system (40). In COPDps, the prevalence of osteoarthritis is estimated at an average of 35.5% (41), which increases the risk of falls, and makes this group of patients more vulnerable to falls than other elderly. Huang et al. (42) examined 55,000 people and estimated the number of hip fracture incidences among COPDs at 649 per 100,000 person-years, which was more than two times the reported data for Non-COPD patients.

Generally speaking, the fall risk factors in individuals can be divided into two categories: patient-related (intrinsic) and environmental hazards (extrinsic) (27). The following section addresses these:

4.1. Intrinsic risk factors

4.1.1. Skeletal Muscle Dysfunction

COPD has a wide range of detrimental effects on muscle function over time, including muscle weakness, atrophy, shift in fiber type, poor oxidative capacity, and mitochondrial dysfunction (43). Many studies have reported a decrease in muscle volume, strength, endurance, and capillary density in these patients (44-46). These changes included quadriceps hamstrings and adductor muscle groups, with a decrease in CSA and mass for the lower torso muscles (18). For distal muscles, such as dorsi flexor and plantar flexor, EMG data also demonstrated weakness, as well as, higher fatigability in these individuals as compared to the control group (47). This problem (tendency to get tired and lose strength) could likely be related to decrease of muscle contractile and oxidative capacity due to impairments in gas exchange and O2 delivery in these patients (21). Lower limb muscle strength is an essential factor for having a proper response to recover from a fall (48), and hence the weakness of these muscles leads to the inability to produce sufficient muscle torque around the lower torso joints, especially at the ankle joint which requires synergistic cooperation and appropriate tendon stiffness (49). Atrophy, or deterioration in muscle

mass, also causes decrease in the functional performance of muscles, which may lead to lack of proper response to external disturbances (50). Some of the muscle problems which occur in COPDp are related to aging and are common in the elderly. However, a study performed by Mathur et al., (51) revealed that the muscle volume, as well as, strength were 25% less in these individuals as compared to an age-gender-matched population. In addition, intramuscular fat amounted to 35% more in the COPDp, indicating a decrease in function and mobility in these patients as compared to their age-matched counterparts.

4.1.2. Impaired activities of daily living:

Numerous studies have shown a significant association between decreased daily activity and increased risk of falls in older adults (52-54). In COPD patients, due to shortness of breath, these patients usually get accustomed to a sedentary lifestyle, leading to a viscous cycle of exacerbation and further diminished activity. The shortness of breath naturally contributes to low levels of ADLs and leads to muscle deconditioning, and eventually, an excess lactate-CO2 production, which exacerbates shortness of breath over time (18). Multiple studies have also reported on the increased inactivity among these patients as compared to age-matched healthy individuals and even to patients with other chronic diseases (55-60). Both qualitative (Self-reported including questionnaires, diaries), as well as quantitative methods (wearable, pedometers, etc.) have been used to monitor the activity of COPDp during the day, revealing less walking time, movement intensity and steps/day.

In 2016, Iwakura et al. (61) examined the relationship between daily activity and various balance characteristics, including gait stability and anticipatory adjustments, quantitatively among COPDp for the first time. The results interestingly revealed that patients' balance was independent of possible confounding factors, such as the loss of muscle strength, and dyspnea, and was more associated with the number of steps per day. On average, these patients performed 4,546 steps, which was almost half of the control group, and less than the threshold of 5000 steps defined for a sedentary lifestyle. Therefore, it seems that balance problems (as a risk factor for fall) and inactivity can intensify each other.

Kawagoshi et al. (62) Examined the walking intensity of COPDp and found that they allocated less time to walk faster (faster than 2 km / h) or slower (at less than 2 km/h) during the day as compared to a healthy control. These patients also demonstrated lower frequency of standing up. They suggested that some of the daily activities, including sitting and standing up from a chair, going up and down stairs, walking at different speeds, were significantly reduced in these patients, which are necessary for proper stability during gait and typically increase anticipatory capabilities in the face of unexpected disturbances. Another study conducted on faller COPDp, stated that the experience of falling leads to a decrease in balance confidence and daily activity (63), yet another factor contributing to the risk of fall. Oliviera et al. (33) also cited balance problems after exacerbation as the reason for the increased incidence of COPD falls, which may be related to inactivity following exacerbation, also reported by Alahmari et al. (64, 65).

4.1.3. Balance deficits

The ability to maintain stability and balance to prevent falls in the elderly is critical. It is not surprising that many studies have identified lower back muscle weakness and balance problems (66, 67) as the most important risk factors for falls (68). Furthermore, in COPD patients, the relationship between different components of balance and falling has been well investigated (29-31). Due to diaphragmatic muscle dysfunction and increased respiratory demand in COPD patients, trunk muscles (EO, IO, and ES) are required to be more active in their secondary role to aid respiration (69). This over-activation increases the stiffness of the trunk, and thus reduces its contribution to maintaining balance (70-72). In a study by Smith et al. (70-72) on post-balance and sway during standing using COP and EMG data, COPDp exhibited decreased trunk maneuverability (and core segments in

general) which was reported as their main cause of balance problems in the ML direction. On the other hand, although increasing respiratory frequency could have caused AP disturbance to their posture, COPD patients overcame this by increasing the hip AP angular motion as a compensatory strategy, and hence the sway value was not significantly different from the control group in the AP direction. ML balance problems in the elderly have been associated with increased fall risk (73). By comparing proprioceptive control strategies between COPDp and a control group, Janssens et al. (74) found that these patients were more likely to rely on the ankle joint (muscle proprioceptive signals), decreasing their reliance on the back during balance control, due to an impaired postural contribution of the inspiratory muscles to trunk stability.

Using the ankle-steered strategy, makes these patients vulnerable to falls, especially during unstable conditions, such as standing on foam. Weakness of the diaphragm, and its impact on trunk and pelvic muscles, cause postural alignment changes in these patients leading to further balance difficulties. Thoracic kyphosis, anterior pelvic tilt and forward head posture are known as the general features of posture in these patients (75-79), which biomechanically causes the displacement of the center of mass and the occurrence of balance problems.

4.1.4. Gait deficit

Gait or walking is reported as the main pre-fall activity leading to hospitalization (80). Poor gait function is considered as a major fall risk factor in COPD patients (27). Several studies have reported slower walking speed in these patients, as compared to healthy individuals (81-83), which may be in part due to weakness of the lower torso muscles, as well as limited daily activity resulting in the reduction of quantity and quality in both soft and hard musculoskeletal tissue compromising the function of gait. In several studies, Yentes et al. (79, 84-86) examined the biomechanical effects of COPD on the gait of these patients. The results showed a decrease in step width and step length and cadence, as well as, an increase in stride time and double support as compared to control. Decreasing the step width reduces the base of support, and thus increases the risk of falls.

Lahousse et al. (87) reported rhythm problems during the gait of 196 COPD patients as compared to healthy individuals regardless of age, sex, exacerbation and dyspnoea. Fallers with COPD had worse rhythm than non-fallers, which is consistent with other studies which reported an association between variables such as rhythm, variability and cognitive decline and risk of falls (88). This group proposed the *gait Rhythm* variable, a novel spatiotemporal gait variable, as a clinical criterion for assessing the risk of falls in patients with respiratory problems, as well as *gait variability*, which reflects system flexibility. Using an accelerometer, Annegarn et al. (89) observed that ML variability in trunk acceleration was higher in COPD patients than controls. They suggested that these individuals may use an active control strategy to compensate for balance disturbances and maintain stability in the AP direction, and therefore have less control over the ML direction, which in turn leads to increased ML variability and increases the risk of falling.

Yentes's et al. (90), further studied he control strategy of COPDp by attempting to answer the important question of why do these patients opt for balance control in the AP direction considering that control in the ML direction increases their risk of falling. By simultaneously examining the dynamic stability (margin of stability (MOS)) and energy expenditure during gait, this study showed that an increase in metabolic power was associated with a decrease in MOS in the AP direction; however, there was no significance correlation with the mean standard deviation or variability in ML direction. Therefore, because these patients are normally at the maximum level of energy consumption due to their increased effort to breathe adequately, it seems that their control system selects a strategy to maintain stability in the AP direction during gait at a normal level, but eventually, their balance problems in the ML direction are shown by increasing the mean and variability of MOS_ML. This study highlights the potential for energy expenditure to be used as an effective measure/determinant for gait problems in COPDp. The results of

the investigation by Smith et al. investigation (70), confirmed these results by examining the sway of COPDp, which reduced trunk involvement in maintaining stability and led to balance problems in the ML direction due to increased muscle stiffness.

The kinematics and kinetics of gait in COPDp have been thoroughly investigated. Yentes et al. (79) examined the lower joints' kinematics and kinetics of gait in COPD patients, as compared to a control group. They found no significant differences, except for the lack of increase in peak dorsiflexion moment in fatigue conditions, as well as a larger ankle joint peak power absorption at mid-stance, which was attributed to weakness of the anterior tibialis. These results reflect that variability in gait spatiotemporal variables is not always sufficiently discriminative between healthy and COPD patients. Other dynamic variables, such as sample entropy and local divergence exponent (91), were therefore used to quantify joint movement patterns in this population. The results demonstrated more restricted (decreased variability of fluctuation, increased predictability) and less consistent organization (decreased local dynamic stability) of lower limb joint movement patterns in COPDp during gait. Lack of sample entropy may indicate that the system was too rigid or too irregular, and consequently has less ability to adapt (such as experiencing an unexpected disturbance). An increased local divergence exponent has already been reported in the elderly in association with the history of falls (92). The study further revealed any changes of speed from self-selected, lead to increased problems of consistency in the organization of the movement patterns of the lower torso joints. They concluded that in general COPDp are unable to respond appropriately when faced with alterations in task demands or environmental disturbances which predispose them to fall.

In addition to kinematics, it is important to consider the kinetics of movement in COPDp, particularly in response to external disturbances. Lower body muscles need to produce enough compensatory torque across the various joints to maintain the COM within the base of support and protect an individual from falling. In COPDp, muscle weakness and motor disorders cause impairment in joint torque production against environmental perturbation, which leads to abnormal fall response and related injuries. Yentes et al. (79) reported a lack of a peak in the dorsiflexion moment only after heel-strike in COPDp gait, and suggested that this abnormality results due weakness in the tibialis anterior. Up to date, and to the best of our knowledge, there are no other investigations on COPDp gait kinetics in other planes aside from the sagittal plane and in response to a sudden perturbation. As falls in the real world are multidirectional, hip muscles need to produce enough joint adductor/abductor moments in the frontal plane to protect from falling in response to perturbation in the medio-lateral direction, this emphasizes the need and importance of 3D kinetics analysis in COPDp. Whether a traditional gait lab with motion capture systems and force plates are used, or novel wearable technology such as sensors and IMUs in conjunction with musculoskeletal models (for example, AnyBody from Aalborg Denmark), such information is significant for a comprehensive 3-D biomechanical analysis which bridges the gap in literature.

4.1.5. Cognitive functioning

4.1.5.1. Cognitive impairment

The brain, which has the highest demand for oxygen than any other organ in the body, can also be vulnerable to the systemic effects of COPD, where cognitive problems are reported in approximately 36% of these patients, or 3 times more than healthy individuals (93). Cognition is the mental ability through which human behavior can be adapted to new situations and/or preference changes (94). Executive functions, language, visuospatial and motor function, memory, attention and social emotions are all domains of human cognition (88).

Major involvement of associative cortical areas, in integration of sensory and executive tasks, explain the link between reduced cognitive capacity and increased risk for falls (27). The prefrontal cortex (PFC) is involved in attentional control and executive functions, including memory, planning, and decision making (95). In COPDp, the structural changes

of PFC that usually occur with aging, are more pronounced than those in the same age group (93). Lower reaction time, delayed memory, and slower information processing speed are among the main problems observed in this group of patients (93). Different studies (96, 97) have reported an increase in the number of road accidents in COPD patients as compared to other elderly groups, which may be attributed to the cognitive disabilities. In addition, other structural changes such as reduced brain perfusion in both the anterior and sub-cortical areas, reduced motor cortex activity, and alterations in the neural structures (white matter lesion, brain atrophy), have also been identified in these patients (98).

In COPDp, researchers have observed a decrease in the ability to perform multiple complex tasks, such as driving or visuo-guided muscle contractions, which require the simultaneous use of cognitive ability and vestibular or visual systems. They suggested that the capacity to perform several simultaneous tasks is impaired due to frontal lobe dysfunction and an associated decrease in attentional capacity (99).

This disorder, which is called Dual-Tasking (Cognitive-Motor) Interference (DTI), also impacts the gait and activity of COPDp. When a person's cognitive ability decreases, the ability of appropriate response (e.g., in the face of obstacles during walking) also decreases, resulting in falls. In COPDp, research has shown that it took longer to perform a functional test with a cognitive task (100). Moreover, delay in responding to task changes has been reported in these individuals, and hence in the face of unexpected environmental disturbances during walking, such as uneven surfaces, slipping or tripping, cognitive and DTI challenges interfere with the needed reaction to maintain balance. In healthy people, even those with cognitive problems, it is possible to partially compensate for delayed responses by increasing muscle activity and producing sufficient joint torque. Muscle weakness, along with delayed reaction and difficulty in selecting the appropriate control strategy in COPDp increases their risk of falling (99).

Research on COPD gait in dual tasking revealed an increase in gait variability along with a decrease in the speed of gait. Both are measures of gait stability impairment, which is closely associated with the increased risk of falls. Furthermore, structural changes in the brain due to the reduced oxygen delivery caused by COPD, in addition to the cognitive impairment, also manifests itself in the form of neural drive to muscle problems. Alexandre et al. (101) showed that knee extensors cannot be optimally driven by the brain while receiving torque control visual feedback (dual task) in COPDp. Therefore, in addition to muscle weakness, higher DTI during muscle force production also increases the risk of falling in these patients, so that both the reaction time and the mental strength/capacity of these patients to external disturbances are disturbed due to problems in the brain caused by COPD.

4.1.5.2. Mental (cognitive) fatigue

Mental fatigue can be defined as a temporary as the temporary inability to maintain optimal cognitive performance resulting from hard mental or physical (102). Almost fifty to seventy percent of COPDp experience mental fatigue in their lives, which disrupts ADLs, reduces the quality of life and may even lead to temporary or permanent hospitalization of these patients (103). Fatigue generally falls into two categories: peripheral (performance) and central (perceived). Peripheral fatigue is identified as changes in objective measures of performance over a defined period of time, i.e. decline in the peak force generated during maximum voluntary muscle contraction, while central fatigue is a progressive failure to voluntarily activate muscles, which can originate at the spinal or supraspinal levels, and one of its components is mental or cognitive (104). This form of fatigue is considered as a psychobiological state, which is typically triggered by prolonged periods of demanding cognitive activities. Research has shown that mental fatigue is associated with changes in the activation of the prefrontal cortex, which is the part of the brain, involved with executive function, and is most likely damaged in COPDp. Therefore, it can be assumed that the mental fatigue most likely has a substantial impact on functional impairment in COPDp (105).

Behrens et al. (106) hypothesized that cognitive fatigue among older adults may lead to changes in gait and postural control, both of which require cognitive processes. In this study, the effect of mental fatigue on gait performance under single- and dual-task conditions was investigated in young and old participants. Questionnaires were used to review cognitive functioning (MMSE), Fall (FES-I)), Fatigue level (MFIS), wakefulness and mood (MDMQ) among the subjects. Variables such as psychophysiological workload, heart rate variability, as well as gait variability were evaluated during the experiment. Their results indicated that mental fatigue causes impaired gait performance (increased variability) during dual-task walking in old adults, and it should therefore be considered as a potential risk factor for the occurrence of falls in the elderly.

Up to date, the effect of mental fatigue on the gait of COPDp has not been studied, although Yentes et al. (79) reported minimal significant differences (only in the ankle joint power and peak moment) under REST and No-rest conditions among these patients. They asked patients to walk on a treadmill at their desired speed until they felt tired in the legs or shortness of breath, and then to perform 5 trips of 10 meters in a row without resting. Although they used this experiment to investigate the effect of fatigue on the gait biomechanics of COPDp, the results did not clearly show the complex relationship between fatigue, gait disorders and falling in this population.

In COPDp, impaired O2 delivery to both the pre-frontal cortex and the active limb muscles contributes significantly to a reduced central motor drive and the development of accelerated peripheral muscle fatigue, respectively (21). Thus, although mental and peripheral fatigues are manifested by separate mechanisms, they are systematically related and can exacerbate each other's effects. Impairment in gas exchange (e.g., decreased arterial O2 pressure) in COPDp due to fatigue, causes decreased cardiac output (21). Although the blood flow is reduced, the breathing effort is increased due to diaphragmatic disorders and absorbs more blood volume, resulting in reduced blood supply to the lower extremities and fatigue-induced movement problems. On the other hand, the simultaneous increase of afferent information from the respiratory/peripheral muscles, along with the decrease of brain capacity due to impaired cerebral oxygenation, constitute the mechanism of central fatigue in COPD individuals. As mentioned, after the manifestation of peripheral fatigue, aggravation of disease or strenuous daily activity (such as high-speed walking, climbing stairs, etc.), even if intermittent, could likely lead to centralized fatigue due to exacerbating respiratory needs and sending increased nerve signals to the brain. During dual tasking, when part of the brain's capacity is already involved, the effects of this form of fatigue will be more severe. In COPDp who are quite vulnerable to mental fatigue incidence, there is likely to be a significant association between gait and falling disorders with this form of fatigue. Examination of mental fatigue in people with COPD can highlight the importance of this phenomenon as a risk factor for falling, which can also disrupt the daily lives of these patients.

 $\textbf{Table 1.} \ \textbf{literature review summary of fall risk factors in COPD patients}.$

author	risk factor	objective	observation
Gosker et al. (50) 2007	Skeletal muscle dysfunction	determining whether vastus lateralis muscle fibre type proportions are associated with COPD disease severity	the relationship was found between skeletal muscle abnormalities and important hallmarks of the disease in severe COPD
Mathur et al. (51) 2008	Skeletal muscle dysfunction	examining the relationship between muscle cross-sectional area (CSA) and volume by use of MRI in healthy and COPD individuals	muscle volume and strength were 25% less in COPDp as compared to an agegender-matched healthy population
Iwakura et al. (61) 2016	Impaired activities of daily living	investigating the association between balance and physical activity measured by an activity monitor in elderly COPD patients	Impairments in balance and reductions in physical activity were observed in the COPD group
Kawagoshi et al. (62) 2013	Impaired activities of daily living	quantifying the walking time and frequency of postural changes in daily life in patients with COPD	the COPD patients' different walking times in daily life are significantly correlated with exercise capacity and dyspnea
Smith et al. (70) 2016	Balance deficits	and trunk muscle activity differs in people with and without COPD before and/or after exercise	severe COPD is associated with impaired ability to recover balance and greater trunk muscle activity during postural challenges
Janssens et al. (74) 2013	Balance deficits	Determine the specific proprioceptive control strategy during postural balance in individuals with COPD and healthy controls	COPDp were more likely to rely on the ankle joint (muscle proprioceptive signals), decreasing their reliance on the back during balance control
Lahousse et al. (87) 2015	Gait deficit	investigating associations of COPD with various gait domains and explored a potential link with falling.	Persons with COPD exhibit worse Rhythm, especially fallers with COPD.
Fallahtafti et al. (90) 2020	Gait deficit	Is dynamic stability different between COPD patients and controls?	Patients with COPD operate at the upper limit of their metabolic reserve due to an increased cost of breathing
Hassan et al. (98) 2019	Cognitive impairment	comparing changes in dorsolateral prefrontal cortex (DLPFC) oxygenated hemoglobin (Δ O2Hb), accuracy of backwards spelling, and decrements in gait velocity during single and dual tasks in patients with COPD and healthy younger and older adults	Decrements in performance during dual tasking highlight the impact of increased cognitive load and need of cognitivemotor interventions to improve cognition and physical function
Heraud et al (99) 2018	Cognitive impairment	Assessing the cognitive and motor performances of patients with COPD and healthy controls with in a dual task walking paradigm	providing evidence of insufficient attentional resources to successfully deal with DT in patients with COPD, and this was expressed through an exaggerated increase in gait variability in DT walking
Paneroni et al. (105) 2020	Mental (cognitive) fatigue	evaluating the impact of exercise training on fatigue, compared with normal care in patients with COPD	No study evaluating performance fatigue was found
Behrens et al. (106) 2018	Mental (cognitive) fatigue	cognitive fatigue among older adults may lead to changes in gait and postural control, both of which require cognitive processes	mental fatigue, induced by sustained cognitive activity, can impair gait performance during dual-task walking in old adults

4.2. Extrinsic risk factors

Slipping and tripping during walking are the most common causes of falls in the aging population and in patients with increased fall risk (80). Their inability to

predictively (before the perturbation) or reactively (after the perturbation) adapt to changes and challenges in the environment causes increased risk of falling in these people. There is a need to physically prepare patients' perspective of falling for situations where unexpected mechanical disturbances to gait/balance could occur. While General exercise interventions (muscle-strengthening + balance exercises) moderately improve falls risk (14–17%), training reactive recovery responses following sudden perturbations (compensatory stepping, counter rotation or grasping actions) during walking may be more taskspecific for falls prevention (80). The comparison of balance responses to external perturbations between COPD and healthy people reveals an association between COPD and fall risk factors. Gait perturbation can be simulated by methods such as: moveable floor platforms, ground surface compliance changes and belt accelerations or decelerations (AP perturbation), and/or moving the treadmill surface to the left and right (ML perturbation) during treadmill walking. The treadmill has advantages over the two other methods, including providing a feasible and less predictable platform (more difficult for participants to anticipate perturbations), in addition to the capacity of applying perturbations in both AP-ML directions and simulating continuous gait (80).

McCrum et al. (80), in their pilot study on 12 COPD patients, find no significant deficits in stability following sudden treadmill-based perturbations in the AP direction in this group. While previous studies demonstrated that COPDp suffer from balance problems in the ML direction (70-72, 90), this study only focused on AP perturbations, in addition to the limited number of subjects. Moreover, the effects of parameters, such as mental and physical fatigue, cognition, medication, history of falling, as well as the other well-known COPD related fall risk factors, were not investigated. The subjects had a FEV1/FVC < 0.70 and were able to complete a six-minute walking test, however, the severity disease of the subjects was not described. It is likely that all these patients were in a relatively good health condition (FEV1/FVC close to 0.7), whereas by subgrouping them based on the GOLD standard (or other methods such as the BODE index), the relationship between the severity of the disease and individuals' ability to maintain stability can be better defined. Further investigations are required to study how COPD affects balance responses to sudden perturbations in the COPDp population in comparison to age-matched healthy controls.

4.3. Hypotheses: Possible causes underlying the increased risk of falls in COPD patients

All the above-mentioned studies have reported serious motor-balance problems in COPD patients and their association with an increased risk of falling in this population. On the other hand, there remains a need to pinpoint the underlying mechanism(s) leading to falls in COPDp and examine the importance of different risk factors, their interrelationships, and association with patients' response to balance after a fall. The question of how much are inactivity, energy expenditure, gait difficulties, and cognitive problems responsible for COPD patients falling remains an open area of research. The following section delineates several of the possible hypotheses based on the literature:

H1- To keep energy expenditure at a normal level, patients with COPD are likely to reduce the intensity of their daily physical activity. Moderate- and vigorous-intensity physical activity requires anticipatory adjustments, gait stability, and transitions. There is a significant relationship between the level of daily life activity intensity and ability of COPDs to regain balance after a sudden external perturbation: As mentioned earlier, the gait of COPDp is typically adapted to maintain stability in the AP direction, albeit at the cost of instability in the ML direction due to the importance of energy expenditure control for COPD patients, which consequently, leads to an increased risk of falling (90). Therefore, this hypothesis assumes that such reduction in energy consumption can cause gait deficits and hence increase the risk of falling. When Hugli et al. (107) monitored the daily activities of COPDp and measured total energy expenditure, they found that, patients with stable COPD had normal daily energy expenditure. It seems that in addition to changes in the amount of daily activity of these patients, the intensity of the activity was also altered to control daily energy

expenditure at a normal level. These changes, on the other hand, resulted in a weak response to external disturbances and hence in frequent fall incidence. This hypothesis is consistent with the results of Kawagoshi et al. (62). They found that the lack of activities with high/low intensities, such as fast/slow walking, going up/down the stairs, and sitting/getting up from a chair in COPDs' daily activities, to be an important problem which reduced the ability of the patients to provide an appropriate postural response when faced with a challenging or unexpected situation. Therefore, controlling (and reducing) energy consumption as an important constraint causes changes in various aspects of movement, as well as, the amount and quality of physical activity of COPDp.

H2- An appropriate modular organization of muscle Synergies is required to maintain postural control under challenging balance conditions. Physical inactivity in COPDp can result in differences in modular control of muscle coordination patterns during walking in this population as compared to healthy individuals. COPD decreases motor control performance in response to an unexpected gait perturbation by affecting the modular organization of muscle activity patterns in these patients: Inactivity affects the status of COPD patients via different mechanisms, such as the well-known cycle of "shortness of breath-inactivity-muscle deconditioning-increased carbon dioxide-shortness of breath". Therefore, a sedentary lifestyle can exacerbate the disease and possibly lead to increased risk of fall and injuries due to the patients' compromised musculoskeletal condition. Several studies have also identified balance deficits as an important risk factor for COPD falls. It can be therefore hypothesized that inactivity weakens the lower body muscles, such as the quadriceps, which synergistically along with the pelvic and abdominal muscles, contribute to postural stability. This leads to an increased trunk stiffness and decreases the trunk's contribution in maintaining balance, which consequently increases the risk of fall and/or the disruption in response to external disturbances.

H3- COPD patients suffer from cognitive impairment. Cognitive function is associated with falls and gait deficits in older adults. There is an association between domain-specific cognitive impairment in COPDp and their gait performance and response to external perturbation: Among all the variables reported by Lahousse et al. (87) regarding the gait of COPDp, only the "rhythm" indicated a significant difference between patient and control groups, as well as fallers and non-fallers. It has already been demonstrated that there is a relationship between memory and gait rhythm (88). Considering that COPD patients are at high risk of cognitive impairment, the relationship between different cognitive domains and gait impairments reported in this population is not studied yet. It can be hypothesized that COPD causes gait rhythm impairment by affecting cognitive ability, in particular memory, which increases the risk of falling in these patients. In addition, the correlation between brain structure/function and performance of gait and balance recovery responses to external perturbations in COPD patients remain controversial and equivocal.

H4- "Mental fatigue" is a risk factor for falling in COPDp and results increased gait variability and instability during dual-tasking: It has been reported that there is a close association between emotional status and movement alteration (108). In COPDp because of decreased cerebral gray matter (109), a potential increase in the risk of depression and anxiety is expected, which in turn, may cause movement alterations/deficits, especially during gait, as well as a higher risk of falling. Furthermore, since proprioceptive afferents emerging from muscles and connective tissue are unable to be adequately processed by the cerebellum in COPDp, iImpaired proprioception is another culprit behind control problems during movement and increased fall incidence in COPDp. Cognitive overloading due to the patients' chronic condition, when their cognitive functioning with aging and reduced oxygen supply to brain are already impaired, further increases the vulnerability of COPDp and their susceptibility to mental fatigue as compared to a normal elderly group, especially during dual-tasking.

H5- COPD patients have different "trunk coordination" relative to lower-body segments while walking and after perturbation, which should be considered as a fall risk factor: Most of reviewed studies considered gait disorders as key to understanding the relationship between the systematic effects of disease and falling in COPDp. Using 3D motion analysis,

Yentes et al. (79) studied the characteristics of gait in COPDp focusing on the kinetics and kinematics of the lower-limb joint (ankle-knee-hip) in the sagittal plane. On the other hand, COPD is mainly an upper-limb disease based on the location of lungs and the pelvis and trunk play an important role in the manifestation of the disease. Moreover, while studies, including this one, focused only on the sagittal plane, hip adductor weakness reported in COPD patients confirms the importance of the incorporation of the frontal plane and the need for a full 3-D biomechanical approach. Although different studies (85, 91) investigated the variability of spatial-temporal parameters of the gait (i.e. step width, step length, cadence, etc.) of COPDp, the majority of these studies focused on end-point variability (the variability of the results of a movement, i.e. step width, length, etc.) and not the working point variability (the variability of segments and joints of a movement). Segmental coordination and coordinative variability are advanced mathematical methods that can be used to analyze a segment (such as the trunk) kinematics relative to other segments (pelvic, shank, etc.) in different anatomical planes (sagittal, frontal, transverse). "Segmental coordination" can help understand the effect of COPD on motor control and how to respond properly to perturbations and prevent injuries during the loss of balance and falls (110). However, we can hypothesize that coordination exercises, such as Tai Chi, can regulate breathing, strengthen upper and lower body musculature, and improve COPDs' trunk coordination, hence improving overall postural stability.

H6- COPD patients with "low back pain" (LBP) have different balance strategies to regain stability after unexpected perturbations, as compated to LBP-free and healthy age-matched controls, because they rely more on ankle rather than back muscles: Diaphragm dysfunction is a major cause of non-specific LBP among COPD individuals (75), where the prevalence is estimated between 41.2 and 69 percent (75). In addition to causing pain and reduced quality of life, LBP is a potential risk factor for falls. Researchers have shown that elderly people with LBP are almost three times more likely to fall as compared with healthy elderly, in addition to exhibiting greater postural sway, longer reaction time, and lower quadriceps strength (78, 111). As previously mentioned, in addition to its respiratory role, the diaphragm also helps the back muscles in maintaining spinal stabilization, and hence its dysfunction causes instability of the vertebrae and leads to LBP over time (69, 75). In a closed biomechanical chain, a decrease in diaphragm efficiency due to weakness, deformation, or fatigue causes the spine to rotate around the T8 level, which in turn causes the pelvic muscles to be stretched, exacerbating the pelvic tilt. Synergistically, the hip joint will rotate in the opposite direction, resulting in a rotation of the femoral-tibial coupling, followed by ankle pronation. Therefore, a diaphragm problem, along with this biomechanical chain of disorders, typically manifest at the ankle joint level, as confirmed by other studies (79). If not treated with proper rehabilitation, this will aggravate the diaphragm problem and further worsen the associated lung disease.

5. Limitations

Roig et.al (26) divided the risk factors of COPDp in two groups of intrinsic and precipitating factors. Intrinsic factors are associated with the physical and psychological status of patients, such as muscle weakness, gait and balance problems, cognitive disorders, etc., while precipitating factors, such as exacerbations and dyspnea, are health-related factors. Extrinsic factors can be added as third group and are associated with environmental factors, such as weak light and slippery conditions, but are not affected directly by the disease. Although both extrinsic and intrinsic factors were discussed in this study, some aspects pertaining to intrinsic risk factors, such as visual deficits and use of an assistive device, were left out due to the lack of sufficient related articles. Furthermore, Roig et al. introduced medication and central nervous system medications (e.g. antidepressants and narcotics) as potential risk factors in these COPDp. High usage of some medications (e.g. antidepressants and narcotics) and sedatives (benzodiazepines), as well as corticosteroids, in these patients can cause some side effects, such as impaired postural control, slowed postural reflexes, muscle weakness and also can increase the risk of fracture (26). All the

aforementioned factors can potentially increase the risk for falling in COPDp. Although, the relationship between the type and number of medications with the risk of falling was not addressed in this study, it is recommended that future research investigate the mechanisms that explain this relationship.

6. Conclusions and Future Directions

This study aimed to 1. analyze the literature to identify and summarize the risk factors of falling in COPDp populations, and 2. investigate the underlying mechanisms by which these risk factors can lead to an increased prevalence of falling. The results confirm that the risk of falling in COPDp is substantially higher, as compared with age-matched healthy individuals, due to well known risk factors, including gait and balance problems, cognitive disorders and reduced daily activity. However, the precise underlying mechanism(s) which lead to increased falls in these patients remain unclear. Based on the literature reviewed here, this study also suggests that low back pain and mental fatigue should be added to the list of potential fall risk factors. This study importantly explored the possibility of a direct correlation between the improvement in performance of COPDp in response to unexpected disturbances and the intensity of daily activities, confirming that the probability of falling may likely decrease by performing daily activities or sports at low and high speeds according to the age and the disease level of these patients. Although up to date no significant differences have been reported in most of the kinetic-kinematic gait variables reported in the literature for COPDp as compared to healthy individuals, this study recommends analyzing the segmental coordination and intramuscular muscle synergy as means to understand the association between the gait challenges associated with the disease and subsequent fall mechanics. Future work is warranted towards understanding the underlying mechanisms leading to falls in COPDp by evaluating the responses to unexpected disturbances during walking in both ML and AP directions and by examining the impact of each of these risk factors on improving these responses.

References

- 1. Wilke J, Mohr L, Tenforde AS, Edouard P, Fossati C, González-Gross M, et al. A pandemic within the pandemic? Physical activity levels have substantially decreased in countries affected by COVID-19. 2020.
- 2. Yates T, Razieh C, Zaccardi F, Rowlands AV, Seidu S, Davies MJ, et al. Obesity, walking pace and risk of severe COVID-19 and mortality: analysis of UK Biobank. International Journal of Obesity. 2021;45(5):1155-9.
- 3. Pauwels RA, Buist AS, Calverley PM, Jenkins CR, Hurd SS. Global strategy for the diagnosis, management, and prevention of chronic obstructive pulmonary disease: NHLBI/WHO Global Initiative for Chronic Obstructive Lung Disease (GOLD) Workshop summary. American journal of respiratory and critical care medicine. 2001;163(5):1256-76.
- 4. De Marco R, Pesce G, Marcon A, Accordini S, Antonicelli L, Bugiani M, et al. The coexistence of asthma and chronic obstructive pulmonary disease (COPD): prevalence and risk factors in young, middle-aged and elderly people from the general population. PloS one. 2013;8(5):e62985.
- 5. Rycroft CE, Heyes A, Lanza L, Becker K. Epidemiology of chronic obstructive pulmonary disease: a literature review. International journal of chronic obstructive pulmonary disease. 2012;7:457.
- 6. Lopez AD, Shibuya K, Rao C, Mathers C, Hansell A, Held L, et al. Chronic obstructive pulmonary disease: current burden and future projections. European Respiratory Journal. 2006;27(2):397-412.
- 7. https://www.who.int/news-room/fact-sheets/detail/chronic-obstructive-pulmonary-disease-(copd).
- 8. Gibson GJ, Loddenkemper R, Lundbäck B, Sibille Y. Respiratory health and disease in Europe: the new European Lung White Book. Eur Respiratory Soc; 2013.
- 9. Terzikhan N, Verhamme KM, Hofman A, Stricker BH, Brusselle GG, Lahousse L. Prevalence and incidence of COPD in smokers and non-smokers: the Rotterdam Study. European journal of epidemiology. 2016;31(8):785-92.
- Vos T, Barber RM, Bell B, Bertozzi-Villa A, Biryukov S, Bolliger I, et al. Global, regional, and national incidence, prevalence, and years lived with disability for 301 acute and chronic diseases and injuries in 188 countries, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. The lancet. 2015;386(9995):743-800.
- 11. Cote C, Chapman KR. Diagnosis and treatment considerations for women with COPD. International journal of clinical practice. 2009;63(3):486-93.
- 12. Wheaton AG, Cunningham TJ, Ford ES, Croft JB. Employment and activity limitations among adults with chronic obstructive pulmonary disease—United States, 2013. MMWR Morbidity and mortality weekly report. 2015;64(11):289.
- 13. Geldmacher H, Biller H, Herbst A, Urbanski K, Allison M, Buist AS, et al. Die Prävalenz der chronisch obstruktiven Lungenerkrankung (COPD) in Deutschland. DMW-Deutsche Medizinische Wochenschrift. 2008;133(50):2609-14.

- 14. Steppuhn H, Kuhnert R, Scheidt-Nave C. 12-month prevalence of known chronic obstructive pulmonary disease (COPD) in Germany. 2017.
- 15. Hurd S. The impact of COPD on lung health worldwide: epidemiology and incidence. Chest. 2000;117(2):1S-4S.
- von Siemens SM, Perneczky R, Vogelmeier CF, Behr J, Kauffmann-Guerrero D, Alter P, et al. The association of cognitive functioning as measured by the DemTect with functional and clinical characteristics of COPD: results from the COSYCONET cohort. Respiratory research. 2019;20(1):1-13.
- 17. Cazzola M, Calzetta L, Bettoncelli G, Cricelli C, Romeo F, Matera MG, et al. Cardiovascular disease in asthma and COPD: a population-based retrospective cross-sectional study. Respiratory medicine. 2012;106(2):249-56.
- 18. Donaldson AV, Maddocks M, Martolini D, Polkey MI, Man WD. Muscle function in COPD: a complex interplay. International journal of chronic obstructive pulmonary disease. 2012;7:523.
- 19. Patten SB, Williams JV. Chronic obstructive lung diseases and prevalence of mood, anxiety, and substance-use disorders in a large population sample. Psychosomatics. 2007;48(6):496-501.
- 20. Jinshil K, Eunok P, Minjeong A. The cognitive impact of chronic diseases on functional capacity in community-dwelling adults. The Journal of Nursing Research. 2019;27(1):1.
- Oliveira MF, Zelt JT, Jones JH, Hirai DM, O'Donnell DE, Verges S, et al. Does impaired O2 delivery during exercise accentuate central and peripheral fatigue in patients with coexistent COPD-CHF? Frontiers in physiology. 2015;5:514.
- 22. Alexandre F, Héraud N, Tremey E, Oliver N, Bourgouin D, Varray A. Specific motor cortex hypoexcitability and hypoactivation in COPD patients with peripheral muscle weakness. BMC pulmonary medicine. 2020;20(1):1-13.
- 23. Førli L, Halse J, Haug E, Bjørtuft Ø, Vatn M, Kofstad J, et al. Vitamin D deficiency, bone mineral density and weight in patients with advanced pulmonary disease. Journal of internal medicine. 2004;256(1):56-62.
- Munn Z, Peters MD, Stern C, Tufanaru C, McArthur A, Aromataris E. Systematic review or scoping review? Guidance for authors when choosing between a systematic or scoping review approach. BMC medical research methodology. 2018;18(1):1-7.
- Agusti A, Noguera A, Sauleda J, Sala E, Pons J, Busquets X. Systemic effects of chronic obstructive pulmonary disease. European Respiratory Journal. 2003;21(2):347-60.
- Roig M, Eng JJ, MacIntyre DL, Road JD, Reid WD. Deficits in muscle strength, mass, quality, and mobility in people with chronic obstructive pulmonary disease. Journal of cardiopulmonary rehabilitation and prevention. 2011;31(2):120.
- 27. Roig M, Eng JJ, Road JD, Reid WD. Falls in patients with chronic obstructive pulmonary disease: a call for further research. Respiratory medicine. 2009;103(9):1257-69.
- 28. Roig M, Eng J, MacIntyre D, Road J, FitzGerald J, Burns J, et al. Falls in people with chronic obstructive pulmonary disease: an observational cohort study. Respiratory medicine. 2011;105(3):461-9.
- 29. Beauchamp M, Hill K, Goldstein R, Janaudis-Ferreira T, Brooks D. Impairments in balance discriminate fallers from non-fallers in COPD. Respiratory medicine. 2009;103(12):1885-91.
- 30. Beauchamp MK, Janaudis-Ferreira T, Parreira V, Romano JM, Woon L, Goldstein RS, et al. A randomized controlled trial of balance training during pulmonary rehabilitation for individuals with COPD. Chest. 2013;144(6):1803-10.
- 31. Beauchamp MK, Sibley KM, Lakhani B, Romano J, Mathur S, Goldstein RS, et al. Impairments in systems underlying control of balance in COPD. Chest. 2012;141(6):1496-503.
- 32. Oliveira CC, Lee AL, McGinley J, Thompson M, Irving LB, Anderson GP, et al. Falls by individuals with chronic obstructive pulmonary disease: A preliminary 12-month prospective cohort study. Respirology. 2015;20(7):1096-101.
- 33. Oliveira CC, Lee AL, McGinley J, Anderson GP, Clark RA, Thompson M, et al. Balance and falls in acute exacerbation of chronic obstructive pulmonary disease: a prospective study. COPD: Journal of Chronic Obstructive Pulmonary Disease. 2017;14(5):518-25
- 34. Hakamy A, Bolton CE, Gibson JE, McKeever TM. Risk of fall in patients with COPD. Thorax. 2018;73(11):1079-80.
- 35. Hellström K, Vahlberg B, Urell C, Emtner M. Fear of falling, fall-related self-efficacy, anxiety and depression in individuals with chronic obstructive pulmonary disease. Clinical rehabilitation. 2009;23(12):1136-44.
- 36. Barbier V, Noppawan C, Loeckx M, Camillo CA, Coosemans I, Muylaert I, et al. Effects of exercise training (ET) in pulmonary rehabilitation programs on balance status and falls in patients with COPD. Eur Respiratory Soc; 2016.
- 37. Brakema E, Van Gemert F, Van der Kleij R, Salvi S, Puhan M, Chavannes N. COPD's early origins in low-and-middle income countries: what are the implications of a false start? NPJ primary care respiratory medicine. 2019;29(1):1-3.
- 38. Johal K, Boulton C, Moran C. Hip fractures after falls in hospital: a retrospective observational cohort study. Injury. 2009;40(2):201-4.
- 39. Kannegaard PN, van der Mark S, Eiken P, Abrahamsen B. Excess mortality in men compared with women following a hip fracture. National analysis of comedications, comorbidity and survival. Age and ageing. 2010;39(2):203-9.
- 40. Haines TP, Nitz J, Grieve J, Barker A, Moore K, Hill K, et al. Cost per fall: a potentially misleading indicator of burden of disease in health and residential care settings. Journal of evaluation in clinical practice. 2013;19(1):153-61.
- 41. Wshah A, Guilcher SJ, Goldstein R, Brooks D. Prevalence of osteoarthritis in individuals with COPD: a systematic review. International journal of chronic obstructive pulmonary disease. 2018;13:1207.
- 42. Huang S-W, Wang W-T, Chou L-C, Chen H-C, Liou T-H, Lin H-W. Chronic obstructive pulmonary disease increases the risk of hip fracture: a nationwide population-based cohort study. Scientific Reports. 2016;6(1):1-7.
- 43. Maltais F, Decramer M, Casaburi R, Barreiro E, Burelle Y, Debigaré R, et al. An official American Thoracic Society/European Respiratory Society statement: update on limb muscle dysfunction in chronic obstructive pulmonary disease. American journal of respiratory and critical care medicine. 2014;189(9):e15-e62.

- 44. Gosselink R, Troosters T, Decramer M. Peripheral muscle weakness contributes to exercise limitation in COPD. American journal of respiratory and critical care medicine. 1996;153(3):976-80.
- 45. Hopkinson NS, Tennant RC, Dayer MJ, Swallow EB, Hansel TT, Moxham J, et al. A prospective study of decline in fat free mass and skeletal muscle strength in chronic obstructive pulmonary disease. Respiratory research. 2007;8(1):1-8.
- 46. Singer J, Yelin EH, Katz PP, Sanchez G, Iribarren C, Eisner MD, et al. Respiratory and skeletal muscle strength in COPD: impact on exercise capacity and lower extremity function. Journal of cardiopulmonary rehabilitation and prevention. 2011;31(2):111.
- 47. Gagnon P, Maltais F, Bouyer L, Ribeiro F, Coats V, Brouillard C, et al. Distal leg muscle function in patients with COPD. COPD: Journal of Chronic Obstructive Pulmonary Disease. 2013;10(2):235-42.
- 48. Porto JM, Cangussu-Oliveira LM, Freire Junior RC, Vieira FT, Capato LL, de Oliveira BGM, et al. Relationship between lower limb muscle strength and future falls among community-dwelling older adults with no history of falls: A prospective 1-year study. Journal of applied gerontology. 2021;40(3):339-46.
- 49. Oliveira AS, Silva PB, Lund ME, Farina D, Kersting UG. Balance training enhances motor coordination during a perturbed sidestep cutting task. journal of orthopaedic & sports physical therapy. 2017;47(11):853-62.
- 50. Gosker HR, Zeegers MP, Wouters EF, Schols AM. Muscle fibre type shifting in the vastus lateralis of patients with COPD is associated with disease severity: a systematic review and meta-analysis. Thorax. 2007;62(11):944-9.
- Mathur S, Takai KP, MacIntyre DL, Reid D. Estimation of thigh muscle mass with magnetic resonance imaging in older adults and people with chronic obstructive pulmonary disease. Physical therapy. 2008;88(2):219-30.
- 52. Rubenstein LZ, Josephson KR. The epidemiology of falls and syncope. Clinics in geriatric medicine. 2002;18(2):141-58.
- 53. Rubenstein LZ, Josephson KR. Falls and their prevention in elderly people: what does the evidence show? Medical Clinics. 2006;90(5):807-24.
- 54. Živković D, Karaleić S, Anđelković I, Aksović N. THE EFFECTS OF PHYSICAL ACTIVITY ON THE BALANCE OF THE ELDERLY. Facta Universitatis, Series: Physical Education and Sport. 2019;16(2):761-81.
- 55. Gimeno-Santos E, Frei A, Steurer-Stey C, De Batlle J, Rabinovich RA, Raste Y, et al. Determinants and outcomes of physical activity in patients with COPD: a systematic review. Thorax. 2014;69(8):731-9.
- 56. López-López L, Torres-Sánchez I, Romero-Fernández R, Granados-Santiago M, Rodríguez-Torres J, Valenza MC, editors. Impact of previous physical activity levels on symptomatology, functionality, and strength during an acute exacerbation in COPD patients. Healthcare; 2018: Multidisciplinary Digital Publishing Institute.
- 57. Matkovic Z, Tudoric N, Cvetko D, Esquinas C, Rahelic D, Zarak M, et al. Easy to perform physical performance tests to identify COPD patients with low physical activity in clinical practice. International Journal of Chronic Obstructive Pulmonary Disease. 2020;15:921.
- 58. Pericleous P, van Staa TP. The use of wearable technology to monitor physical activity in patients with COPD: a literature review. International journal of chronic obstructive pulmonary disease. 2019;14:1317.
- 59. Vorrink SN, Kort HS, Troosters T, Lammers J-WJ. Level of daily physical activity in individuals with COPD compared with healthy controls. Respiratory research. 2011;12(1):1-8.
- 60. Watz H, Pitta F, Rochester CL, Garcia-Aymerich J, ZuWallack R, Troosters T, et al. An official European Respiratory Society statement on physical activity in COPD. Eur Respiratory Soc; 2014.
- 61. Iwakura M, Okura K, Shibata K, Kawagoshi A, Sugawara K, Takahashi H, et al. Relationship between balance and physical activity measured by an activity monitor in elderly COPD patients. International journal of chronic obstructive pulmonary disease. 2016;11:1505.
- 62. Kawagoshi A, Kiyokawa N, Sugawara K, Takahashi H, Sakata S, Miura S, et al. Quantitative assessment of walking time and postural change in patients with COPD using a new triaxial accelerometer system. International Journal of Chronic Obstructive Pulmonary Disease. 2013;8:397.
- 63. Birinci T, Kısa EP, Akıncı B, Kuran Aslan G, Kıyan E. The investigation of falls and balance from the perspective of activities of daily living in patients with COPD. COPD: Journal of Chronic Obstructive Pulmonary Disease. 2021;18(2):147-56.
- 64. Alahmari AD, Kowlessar BS, Patel AR, Mackay AJ, Allinson JP, Wedzicha JA, et al. Physical activity and exercise capacity in patients with moderate COPD exacerbations. European Respiratory Journal. 2016;48(2):340-9.
- 65. Alahmari AD, Patel AR, Kowlessar BS, Mackay AJ, Singh R, Wedzicha JA, et al. Daily activity during stability and exacerbation of chronic obstructive pulmonary disease. BMC pulmonary medicine. 2014;14(1):1-8.
- 66. Davoudi M, Shokouhyan SM, Abedi M, Meftahi N, Rahimi A, Rashedi E, et al. A practical sensor-based methodology for the quantitative assessment and classification of chronic non specific low back patients (NSLBP) in clinical settings. Sensors. 2020:20(10):2902
- 67. Shokouhyan SM, Davoudi M, Hoviattalab M, Abedi M, Bervis S, Parnianpour M, et al. Linear and Non-linear Dynamic Methods Toward Investigating Proprioception Impairment in Non-specific Low Back Pain Patients. Frontiers in bioengineering and biotechnology. 2020;8:584952.
- 68. Loughran KJ, Atkinson G, Beauchamp MK, Dixon J, Martin D, Rahim S, et al. Balance impairment in individuals with COPD: a systematic review with meta-analysis. Thorax. 2020;75(7):539-46.
- 69. Wallden M. The diaphragm-more than an inspired design. Journal of bodywork and movement therapies. 2017;21(2):342-9.
- 70. Smith MD, Chang AT, Hodges PW. Balance recovery is compromised and trunk muscle activity is increased in chronic obstructive pulmonary disease. Gait & posture. 2016;43:101-7.
- 71. Smith MD, Chang AT, Seale HE, Walsh JR, Hodges PW. Balance is impaired in people with chronic obstructive pulmonary disease. Gait & posture. 2010;31(4):456-60.

- 72. Smith MD, Harvey EH, van den Hoorn W, Shay BL, Pereira GM, Hodges PW. Out-patient pulmonary rehabilitation improves medial-lateral balance in subjects with chronic respiratory disease: proof-of-concept study. Respiratory care. 2016;61(4):510-20.
- 73. Maki BE, Holliday PJ, Topper AK. A prospective study of postural balance and risk of falling in an ambulatory and independent elderly population. Journal of gerontology. 1994;49(2):M72-M84.
- 74. Janssens L, Brumagne S, McConnell AK, Claeys K, Pijnenburg M, Burtin C, et al. Proprioceptive changes impair balance control in individuals with chronic obstructive pulmonary disease. PLoS one. 2013;8(3):e57949.
- 75. Bordoni B, Marelli F, Morabito B, Sacconi B, Caiazzo P, Castagna R. Low back pain and gastroesophageal reflux in patients with COPD: the disease in the breath. International journal of chronic obstructive pulmonary disease. 2018;13:325.
- 76. Lee AL, Goldstein RS, Chan C, Rhim M, Zabjek K, Brooks D. Postural deviations in individuals with chronic obstructive pulmonary disease (COPD). Canadian Journal of Respiratory, Critical Care, and Sleep Medicine. 2018;2(2):61-8.
- 77. Muhammed A, Moiz JA, Singla D, Ali MS, Talwar D. Postural abnormalities in phenotypes of chronic obstructive pulmonary disease. Brazilian journal of physical therapy. 2020;24(4):325-32.
- 78. Rosa N, Queiroz BZ, Lopes RA, Sampaio NR, Pereira DS, Pereira LS. Risk of falls in Brazilian elders with and without low back pain assessed using the Physiological Profile Assessment: BACE study. Brazilian journal of physical therapy. 2016;20:502-9.
- 79. Yentes JM, Schmid KK, Blanke D, Romberger DJ, Rennard SI, Stergiou N. Gait mechanics in patients with chronic obstructive pulmonary disease. Respiratory research. 2015;16(1):1-13.
- 80. McCrum C, Gerards MH, Karamanidis K, Zijlstra W, Meijer K. A systematic review of gait perturbation paradigms for improving reactive stepping responses and falls risk among healthy older adults. European review of aging and physical activity. 2017;14(1):1-11.
- 81. Butcher SJ, Meshke JM, Sheppard MS. Reductions in functional balance, coordination, and mobility measures among patients with stable chronic obstructive pulmonary disease. Journal of Cardiopulmonary Rehabilitation and Prevention. 2004;24(4):274-80.
- 82. Ilgin D, Ozalevli S, Kilinc O, Sevinc C, Cimrin AH, Ucan ES. Gait speed as a functional capacity indicator in patients with chronic obstructive pulmonary disease. Annals of thoracic medicine. 2011;6(3):141.
- 83. Zago M, Sforza C, Bonardi DR, Guffanti EE, Galli M. Gait analysis in patients with chronic obstructive pulmonary disease: a systematic review. Gait & posture. 2018;61:408-15.
- 84. Yentes JM, Denton W, Samson K, Schmid KK, Wiens C, Rennard SI. Energy efficient physiologic coupling of gait and respiration is altered in chronic obstructive pulmonary disease. Acta Physiologica. 2019;225(4):e13217.
- 85. Yentes JM, Rennard SI, Schmid KK, Blanke D, Stergiou N. Patients with chronic obstructive pulmonary disease walk with altered step time and step width variability as compared with healthy control subjects. Annals of the American Thoracic Society. 2017;14(6):858-66.
- 86. Yentes JM, Sayles H, Meza J, Mannino DM, Rennard SI, Stergiou N. Walking abnormalities are associated with COPD: an investigation of the NHANES III dataset. Respiratory medicine. 2011;105(1):80-7.
- 87. Lahousse L, Verlinden VJ, van der Geest JN, Joos GF, Hofman A, Stricker BH, et al. Gait patterns in COPD: the Rotterdam Study. European Respiratory Journal. 2015;46(1):88-95.
- 88. Amboni M, Barone P, Hausdorff JM. Cognitive contributions to gait and falls: evidence and implications. Movement disorders. 2013;28(11):1520-33.
- 89. Annegarn J, Spruit MA, Savelberg HH, Willems PJ, Bool Cvd, Schols AM, et al. Differences in walking pattern during 6-min walk test between patients with COPD and healthy subjects. PloS one. 2012;7(5):e37329.
- 90. Fallahtafti F, Curtze C, Samson K, Yentes JM. Chronic obstructive pulmonary disease patients increase medio-lateral stability and limit changes in antero-posterior stability to curb energy expenditure. Gait & posture. 2020;75:142-8.
- 91. Liu W-Y, Schmid KK, Meijer K, Spruit MA, Yentes JM. Subjects With COPD Walk With Less Consistent Organization of Movement Patterns of the Lower Extremity. Respiratory care. 2020;65(2):158-68.
- 92. Lockhart TE, Liu J. Differentiating fall-prone and healthy adults using local dynamic stability. Ergonomics. 2008;51(12):1860-72.
- 93. Andrianopoulos V, Gloeckl R, Vogiatzis I, Kenn K. Cognitive impairment in COPD: should cognitive evaluation be part of respiratory assessment? Breathe. 2017;13(1):e1-e9.
- 94. https://www.cambridgecognition.com/blog/entry/what-is-cognition.
- 95. Yuan P, Raz N. Prefrontal cortex and executive functions in healthy adults: a meta-analysis of structural neuroimaging studies. Neuroscience & Biobehavioral Reviews. 2014;42:180-92.
- 96. Prior TS, Troelsen T, Hilberg O. Driving performance in patients with chronic obstructive lung disease, interstitial lung disease and healthy controls: a crossover intervention study. BMJ open respiratory research. 2015;2(1):e000092.
- 97. Schou L, Østergaard B, Rasmussen LS, Rydahl-Hansen S, Phanareth K. Cognitive dysfunction in patients with chronic obstructive pulmonary disease–a systematic review. Respiratory medicine. 2012;106(8):1071-81.
- 98. Hassan SA. Prefrontal Cortex Activity during Dual Task Performance in Patients with Chronic Obstructive Pulmonary Disease: University of Toronto (Canada); 2019.
- 99. Heraud N, Alexandre F, Gueugnon M, Davy C, Tremey E, Oliver N, et al. Impact of chronic obstructive pulmonary disease on cognitive and motor performances in dual-task walking. COPD: Journal of Chronic Obstructive Pulmonary Disease. 2018;15(3):277-82.
- 100. Morlino P, Balbi B, Guglielmetti S, Giardini M, Grasso M, Giordano C, et al. Gait abnormalities of COPD are not directly related to respiratory function. Gait & posture. 2017;58:352-7.

- 101. Alexandre F, Heraud N, Oliver N, Varray A. Cortical implication in lower voluntary muscle force production in non-hypoxemic COPD patients. PLoS One. 2014;9(6):e100961.
- 102. Marcora SM, Staiano W, Manning V. Mental fatigue impairs physical performance in humans. Journal of applied physiology. 2009;106(3):857-64.
- 103. Spruit MA, Vercoulen JH, Sprangers MA, Wouters EF. Fatigue in COPD: an important yet ignored symptom. The lancet respiratory medicine. 2017;5(7):542-4.
- 104. Grobe S, Kakar RS, Smith ML, Mehta R, Baghurst T, Boolani A. Impact of cognitive fatigue on gait and sway among older adults: A literature review. Preventive medicine reports. 2017;6:88-93.
- 105. Paneroni M, Vitacca M, Venturelli M, Simonelli C, Bertacchini L, Scalvini S, et al. The impact of exercise training on fatigue in patients with chronic obstructive pulmonary disease: a systematic review and meta-analysis. Pulmonology. 2020;26(5):304-13.
- 106. Behrens M, Mau-Moeller A, Lischke A, Katlun F, Gube M, Zschorlich V, et al. Mental fatigue increases gait variability during dual-task walking in old adults. The Journals of Gerontology: Series A. 2018;73(6):792-7.
- 107. Hugli O, Schutz Y, Fitting J-W. The daily energy expenditure in stable chronic obstructive pulmonary disease. American journal of respiratory and critical care medicine. 1996;153(1):294-300.
- 108. Bordoni B, Simonelli M. Chronic Obstructive Pulmonary Disease: Proprioception Exercises as an Addition to the Rehabilitation Process. Cureus. 2020;12(5).
- 109. Wang W, Wang P, Li Q, Peng Z, Wang X, Wang G, et al. Alterations of grey matter volumes and network-level functions in patients with stable chronic obstructive pulmonary disease. Neuroscience letters. 2020;720:134748.
- 110. Glazier A. Measuring coordination and variability in coordination. Movement system variability. 2006:167-81.
- 111. Ge L, Wang C, Zhou H, Yu Q, Li X. Effects of low back pain on balance performance in elderly people: a systematic review and meta-analysis. European review of aging and physical activity. 2021;18(1):1-10.